

Citation for published version: Toan, TD, Meng, M, Lam, SH & Wong, YD 2022, 'Multi-Stage Fuzzy Logic Controller for Expressway Traffic Control During Incidents', *Transportation Engineering Journal of ASCE*, vol. 148, no. 6, 04022027. https://doi.org/10.1061/JTEPBS.0000679

DOI: 10.1061/JTEPBS.0000679

Publication date: 2022

Document Version Peer reviewed version

Link to publication

This material may be downloaded for personal use only. Any other use requires prior permission of the American Society of Civil Engineers. This material may be found at https://ascelibrary.org/doi/full/10.1061/JTEPBS.0000679

University of Bath

Alternative formats

If you require this document in an alternative format, please contact: openaccess@bath.ac.uk

General rights

Copyright and moral rights for the publications made accessible in the public portal are retained by the authors and/or other copyright owners and it is a condition of accessing publications that users recognise and abide by the legal requirements associated with these rights.

Take down policy If you believe that this document breaches copyright please contact us providing details, and we will remove access to the work immediately and investigate your claim.

1

2

Multi-Stage Fuzzy Logic Controller for Expressway

Traffic Control During Incidents

3	Trinh Dinh Toan ^{1*} , Meng Meng ² , Soi Hoi Lam ³ , Yiik Diew Wong ⁴
4	¹ Doctor, Department of Transportation Engineering, Thuyloi University, 175 Tay Son, Dong Da, Hanoi, Viet Nam.
5	* Corresponding author, Email: <u>trinhdinhtoan@tlu.edu.vn</u>
6	² Doctor, School of Management, University of Bath, Bath, BA2 7AY, UK. Email: mengmengbjtu@gmail.com
7	³ Adjunct Professor, Department of Civil and Environmental Engineering, University of Macau, Avenida da Universidade,
8	Taipa, Macau, China. Email: <u>soihoi.lam@gmail.com</u>
9	⁴ Associate Professor, School of Civil and Environmental Engineering, Nanyang Technological University Singapore, 50
10	Nanyang Avenue, Singapore 639798, Singapore. Email: <u>CYDWONG@ntu.edu.sg</u>

11 Abstract: In this research study, a multi-stage Fuzzy Logic Controller (MS-FLC) is developed for traffic control for incident 12 management on expressways. The MS-FLC serves as the traffic operator's decision-making support tool at the operational level. 13 The MS-FLC gathers real-time traffic and incident data in order to analyze and predict traffic conditions as well as to suggest 14 alternative control measures to the traffic operator in the form of linguistic expressions. The MS-FLC is embedded in a traffic 15 simulator controller (TSC) prototype and is evaluated by comparing its performance with no control scenario and ALINEAQ, a 16 popular local ramp control algorithm, across several incident scenarios in a simulation environment. In general, the MS-FLC 17 outperforms ALINEA\Q with respect to global objectives. In particular, whereas the ALINEA\Q algorithm favors the mainline, 18 the MS-FLC algorithm significantly improves mainline travel conditions while substantially reduces ramp queues. It is concluded 19 that, if properly designed the MS-FLC serves as a robust tool for traffic control on expressways under incident conditions.

20 Keywords: Multi-stage fuzzy logic controller; Incident management; Traffic control; Traffic simulator; Ramp metering.

21 Introduction

22 Traffic congestion is a serious and widespread problem in many cities throughout the world. Congestion can be divided into 23 two types: recurring and non-recurring congestion. Congestion management on expressways, which is characterized by time-24 critical constraints, should be enhanced by employing effective real-time control measures to improve traffic conditions. For 25 real-time traffic control, various approaches have been developed, including analytical optimization and automatic control. 26 The analytical optimization approach forecasts the current state of traffic systems based on certain assumptions about system 27 dynamics and behavior, and projects the current network conditions into the future state (Ma et al. 2016; Luan et al. 2018). Mathematical models are usually quite sophisticated and computationally expensive in order to provide systematic solutions 28 29 thus they can hardly meet real-time requirements. The automatic control approach, as opposed to the analytical optimization approach, has the ability to classify enormous patterns of input data in order to describe the behavior of measurable processes
(Simoni and Claudel 2017; Hashemi and Abdelghany 2018; Wang et al. 2018; Lidbe et al. 2019). The technique, on the other
hand, does not include an explanation tool to assist operators in determining appropriate control actions. To address this issue,
a Decision Support System (DSS) is required to make better use of available data, information, and knowledge to improve
the quality of the control decision-making process.

Traffic control is a multivariable problem. The control decision-making process progresses from a low to high degree of abstraction, that is, from data to information to knowledge. For complicated situations where there is a need to evaluate the current traffic situation and to anticipate the future state for determining control actions, the control decision-making process should ideally be stratified into a number of stages where the decision-making logic is executed sequentially from one stage to the next.

Traffic control decision-making is decision-making in the face of uncertainty. Imprecise data measurement, approximate information reasoning, uncertain forecasting of future traffic conditions, and imprecise human perception are all factors that contribute to the unpredictability of traffic control. Because it entails using many forms of traffic and incident data to arrive at control judgments under critical-time restrictions, traffic control in incident scenarios is even more uncertain and critical. Due to the complicated, important, and uncertain nature, an effective traffic control strategy during incidents often relies on techniques that deal efficiently with problems of uncertainty and imprecision.

Fuzzy logic has an attractive capability to deal with uncertainty problems. With the help of fuzzy sets, the vagueness and uncertainties of the real world are handled smoothly. The key motivations behind the application of fuzzy logic for traffic control rest on the following advantages: (i) the linguistic expressions are general and easy to be perceived by the traffic operator, which is important for a decision support system; (ii) the transition from one fuzzy set to another is gradual, representing continuity in human perception; and (iii) the capability to combine several input quantities to provide a single output for the traffic operator to make a control decision (Toan 2008; Toan and Wong 2021).

52 Support Vector Machine (SVM) is a family of machine learning algorithms. SVM possesses a good generalization 53 capability, computational efficiency, and is very robust in high dimensions (Toan and Truong 2021). In traffic engineering, 54 SVM has been successfully applied in many domains, including SVM real-time incident detection (Motamed 2016; Xiao et 55 al. 2013; Motamed and Machemehl 2014), and traffic flow prediction (Yuanyuan and Weixiang 2018; Cai et al. 2018; Luo 56 et al. 2019; Toan and Truong 2021). Short-term prediction of traffic flow is crucial for real-time traffic control.

57 In this study, a multi-stage Fuzzy Logic Controller (MS-FLC) is developed for traffic control for incident management 58 on expressways. The MS-FLC serves as the traffic operator's decision-making support tool at the operational level. The MS-59 FLC gathers real-time traffic and incident data in order to analyze and predict traffic situations, as well as to suggest 60 alternative control methods to the traffic operator in the form of linguistic expressions. Given these functions, the decisionmaking support under these situations typically includes semi-structured decisions (Toan, 2008) that employ both structured modules for data collection, data analysis, and information processing, and non-structured component to help the operators when confronted with qualitative type of decisions. Thus, for the MS-FLC to execute in its totality, SVM is employed as a subset of MS-FLC model for short-term traffic flow prediction for anticipation of incident related traffic condition. Given the anticipated traffic, the MS-FLC calculates the signal settings at the ramp entrance once the operator selects the control measure. The MS-FLC is evaluated in the case study in the "Model evaluation" section. Herein, a traffic simulator controller (TSC) prototype was designed and evaluated across several incident scenarios in a simulation environment.

The remainder of this paper is organized as follows: Section 2 reviews the fundamental concepts and previous works on applications of fuzzy logic systems for expressway traffic control. Section 3 presents the methodology of the MS-FLC that includes the rule-base formulation and the structure of the MS-FLC. Section 4 presents the evaluation results of the MS-FLC, sensitivity analysis, and proposed extension of the MS-FLC for corridor-wide control. Finally, Section 5 summarizes the findings from this research and draws the conclusions.

73 Literature Review

The use of control devices such as traffic lights to regulate the number of cars entering the expressway in order to meet operational objectives such as balancing traffic demand and capacity on the mainline is known as expressway ramp traffic control. Measurable traffic characteristics such as reduced travel time, higher operational speed, or increased throughput have typically been used to evaluate the benefits of ramp control (Zhang et al. 2001). Ramp metering is used to regulate the rate at which traffic can enter an expressway.

Ramp metering control is classified into fixed-time and traffic-responsive strategies (Zhong et al 2014). In fixed-time 79 80 strategy the ramp rates are calculated off-line for various times of the day using the available historical data. Given its static 81 nature, fixed-time strategy may cause either under- or over-utilization of the expressway's mainline. Traffic-responsive ramp 82 metering, on the other hand, adjusts the ramp control in response to the real-time traffic conditions on the mainline and the 83 ramp during the metering period. The adjustment is conducted either in the reactive manner or proactive manner (Toan 2008; Zhong et al 2014). The former adjusts the ramp metering rates using real-time measurements in order to maintain a pre-84 85 specified value of the expressway traffic conditions, while the latter attempts to improve the traffic conditions based on traffic 86 variables anticipated for a certain time horizon. In terms of network topology, ramp metering strategies can be classed as 87 local or coordinated schemes (Zhang et al. 2001; Zhao et al 2016). Local strategy makes use of local measurements to adjust 88 ramp metering rates, whereas coordinated strategy considers a coordination of several controllers in an expressway corridor. 89 The latter utilizes data to simultaneously calculate ramp flows for all controlled ramps within the corridor. Because more 90 extensive information is used and more robust control action is coordinated, this may give possible system-wide gains above

local ramp metering. When there is local congestion, local control is appropriate. Coordinated control should be considered
 if congestion is widespread in different sections of the expressway corridor.

93 Previous research has shown that under recurring traffic congestion, local ramp metering performs compatibly as 94 coordinated approaches, and that local ramp control is the most direct and is an effective strategy to relieve expressway 95 congestion (Papageorgiou et al. 2003). Nonetheless, in the presence of many bottlenecks on the expressway, non-recurrent 96 congestions, or limited ramp storage capacity, coordinated ramp metering systems are often more efficient than local ramp 97 metering strategies (Zhong et al 2014). However, determining whether a ramp metering should be coordinated is not 98 straightforward and is reliant on network topology, background congestion level, and the queue management policies. Rather 99 than launching a complete system, a gradual ramp control strategy should be considered, with priority given to the areas with 100 the largest risk of disrupting traffic flow. However, according to Papageorgiou et al. (1991), the employment of advanced 101 algorithms does not always result in performance enhancement. A local ramp control algorithm ALINEA was tested against 102 coordinated control algorithm METALINE on the Boulevard Peripherique in Paris, using a macroscopic traffic model. The 103 results showed that under normal conditions, both ALINEA and METALINE control systems produced nearly the same 104 results, and the METALINE was only slightly better than the ALINEA in the event of an unforeseen incident due to more 105 comprehensive information.

A fuzzy logic system (FLS) is a non-linear mapping of input to the output universe of discourse using fuzzy logic 106 107 principles. FLS is an attractive approach in handing uncertainty problems. There has been a great deal of works for various 108 applications in traffic engineering such as incident management (Lawrence and Huang 2006; Hatri and Boumhidi 2018; 109 Hawas et al. 2020; Tarig et al. 2020), route choice (Arslan and Khisty 2005; Dhulipala et al. 2017; Bhandari and Cho 2019), 110 safety analysis (Imprialou et al. 2014; Ali et al. 2017; Chowdhury and O'Sullivan 2018), and so on. In the aforementioned 111 applications, FLS in general has delivered promising results. For knowledge representation, many researchers have 112 investigated the rule-based reasoning system for traffic management and control (Toan and Lam 2005; Memon et al. 2015, 113 2016; Yan et al. 2018; Tariq et al. 2020). In the rule-based reasoning system, the knowledge is represented in the form of 114 condition-action pairs: IF conditions (premises) are met, THEN actions (conclusions) are carried out. There are two types of 115 rules: regular rules that evaluate state and control variables using crisp sets, and fuzzy rules that use fuzzy sets. The primary 116 distinction between regular and fuzzy rules is that fuzzy rules allow for partial set membership and a progressive transition 117 from one fuzzy set to the next. The problem-solving capability of fuzzy rules is more competent, thus fuzzy rules are more 118 suitable for complex situations.

Traffic control is one of the earliest applications of FLSs in traffic engineering (Toan and Wong 2021; Chen et al. 2021).
Attempts have been made in this area to use a fuzzy logic technique to improve control at signalized junctions. Pappis and
Mamdani (1977) were the first to use fuzzy logic theory to control traffic at a single signalized intersection. Nakatsuyama et

122 al. (1983), Sasaki and Akiyama (1987, 1988), and others have since made significant contributions to fuzzy logic applications 123 in traffic engineering. Zhan and Prevedouro (2011) introduced a fuzzy logic-based methodology for determining the level of 124 service (LOS) at signalized intersections. The LOS thresholds were replaced with fuzzy values, and fuzzy inferences were 125 used to integrate key factors in order to create a composite LOS measure. The results demonstrated that using fuzzy logic to 126 assess user perceptions of signalized intersection LOS is a viable alternative. Collotta et al. (2015) introduced a traffic signal 127 dynamic control system with multiple fuzzy logic controllers, each handling vehicle turning movements, allowing real-time traffic monitoring. The results showed the system outperformance with considerable reduction of vehicle waiting times. 128 129 Using a formal description of traffic control on crossroads, Yusupbekov et al. (2015) proposed adaptive fuzzy-logic traffic 130 control systems. The results demonstrated that the synthesized adaptive fuzzy control system was robust and capable of 131 directing road traffic over a wide range of parameters. More references on previous literatures in using fuzzy logic for traffic 132 control can be seen in Taylor and Meldrum (2000), Zaied and Al Othman (2011), and Collotta et al. (2015), Kalinic and Krisp 133 (2019), and Tariq et al. (2020).

134 There have been variety of applications of multi-stage fuzzy logic for traffic control. Ge (2014) presented a two-stage traffic signal control method. The first stage calculates traffic urgency degree for all red phases, the second stage determines 135 136 green delay of the current green phase using fuzzy inference. The comparisons were made with pre-timed controller and fuzzy logic controller. The results showed that fuzzy control had a better effect on traffic urgency than pre-timed control and 137 138 common fuzzy control. Based on the Takagi-Sugeno type FLC algorithm, Xu et al. (2013) proposed an efficient local ramp 139 metering approach. The resulting parameters are fine-tuned by particle swarm optimization and microscopic traffic 140 simulations with PARAMICS. Simulation studies show that a balance between traffic on the freeway mainstream and on-141 ramp link has been achieved; Hawas et al. (2019) proposed formulation of a multistage fuzzy-logic model (FLM) for incident 142 detection and management of traffic signals in urban traffic networks. Three distinct non-linear regression models were 143 utilized to find the resilient incident detection and traffic management parameters that are most likely to reduce total network 144 travel time. Other studies on merits of applications for FLC for traffic control are summarized in Yusupbekov et al. (2015), 145 Collotta et al. (2015), and Pandey et al. (2017).

Previous research has taken advantage of fuzzy logic's advantages in dealing with multi-variable traffic control problems, and the results have been promising. Earlier research has shown that in complex situations where it is necessary to analyze available data and information in order to understand the current problem and predict what might happen before proposing a control action, the rules must be executed sequentially according to a decision-making logic. Another reason is that the number of rules increases exponentially as the number of variables increases, thus for a complicated multi-variable control problem the rule base becomes too cumbersome to handle effectively in a single stage, but a multi-stage structure can handle much better. To tackle such complex multi-variable control problems, this research represents the decision-making process by a three-stage control architecture, known as the MS-FLC: output variables from preceding stage are used as input variables to the next stage. The decision-making process in MS-FLC during incident (as presented in the Methodology section) serves to reduce the problem complexity and thereby improves the overall system performance.

In summary, while there has been a lot of work done in the area of fuzzy logic traffic management, the majority of the control applications have been reactive. Little effort has been devoted to traffic control for incident management following MS-FLC approach. Essential issues such as the evaluation of the current traffic situation and anticipation of the immediate incident condition have not been adequately explored, and a systematic procedure in deriving control decisions in the event of an incident have not been adequately addressed.

This research study develops a MS-FLC for expressway traffic control during incidents. The MS-FLC design targets 161 162 application for corridor-wide control for traffic management under both recurrent and non-recurrent congestion. Since the 163 MS-FLC is a highly non-linear system with complex stability behavior, and using the MS-FLC model for corridor-wide 164 management necessitates a significant amount of model calibration effort, the authors propose an incremental development 165 roadmap. Before extending to a corridor-wide control, the MS-FLC is initially built and its performance evaluated using a local ramp control technique, as well as the model's performance sensitivity analysis. Herein, the main focus is on the 166 167 development and assessment of the MS-FLC performance for local ramp control in comparison to competing control algorithms. In the last section, an overall model architecture for corridor-wide control is described. Due to the rarity of off-168 169 ramp control, the phrase "ramp control" in this study refers to on-ramp control.

170 Methodology

171 Overall Framework of the MS-FLC

Fig. 1 describes the proposed architecture of the MS-FLC for incident management. The model reflects a complex sequential structure of the decision-making logic for the multi-variable traffic control problem. The rule base in the MS-FLC consists of 3 stages: (i) incident traffic evaluation; (ii) predicted incident condition; and (iii) recommendation of control action. The rules in the first stage need to be executed first to give results to the second stage. The second stage uses the output from the first stage as its internal input, and external inputs from traffic forecasting. Similarly, the third stage employs both internal and external inputs to provide output in the form of control actions.

178 Stage 1: Evaluation of Current States of Traffic during Incidents

The objective of this stage is to evaluate the current state of traffic in the event of an incident. The traffic state is prescribed by three principal quantities: congestion level (CL), congestion mobility, and congestion status. The congestion level reflects the severity of traffic, estimated by traffic speed and density. The congestion mobility determines the dynamics of the congestion, quantified by traffic speeds. The congestion status refers to the existence and magnitude of queue lengths on expressways. The congestion mobility and congestion status specifically deal with the heavy congestion category. Each component (rule block) requires various treatments in the subsequent stages. If the congestion problem is critical, immediate control measures must be made, and the rules in stage 3 will be executed. By contrast, if the traffic congestion is not yet critical, the system proceeds with traffic forecasting module and rules in the second stage will be fired. The rules in this stage can be categorized as fact-state rules since the reasoning logic uses numerical data to estimate the state of traffic.



188

189

Fig. 1. Conceptual model of MS-FLC for incident-related traffic control

190 Stage 2: Prediction of Incident Traffic Conditions

Predicting short-term traffic conditions is critical to any proactive traffic control scheme's success. The key to anticipating traffic and incident conditions is to predict short-term traffic variables. The second stage, employing short-term traffic prediction advanced traffic forecasting technique for traffic variable predictions and fuzzy logic for data processing, continues to anticipate traffic and incident conditions in the immediate time interval based on the results of the previous stage. The rules in this stage are typically state-to-state rules, since the reasoning sequence infers the future state from the current state using external variables from the traffic-forecasting module.

197 Stage 3: Recommendation of Control Measures

198 The outputs from stages 1 and 2 will be utilized to assess the strength of the necessary control intervention (no control,

199 moderate, strong, and very strong control levels), after which an appropriate control approach will be advised based on the 200 results. Based on the estimated control intervention and the availability of control facilities, the control strategy rule block 201 presents a broad view of alternative control solutions. If concrete control actions are translated, the traffic operator may 202 choose a local or corridor-wide control strategy. Local ramp control, for example, considers ramp traffic and VMS display; 203 corridor-wide control is divided into coordinated ramp control, which coordinates numerous ramp metering controls, and integrated control, which incorporates ramp control as well as VMS diversion directives. The FLC system's outputs are 204 205 defuzzified to provide crisp values. The rules in stage 3 apply to both the strategic (for intervention level, control strategy) 206 and operational levels, as based on the reasoning process (for control settings). Control action rules are essentially state-207 action rules for the given input-output mapping.

208 Rule Base Architecture

209 Given the prescribed relationships, the rules in the proposed MS-FLC can be expressed in the general form:

$$Y = f(X, U) \tag{1}$$

where X is the vector of input variables, U is the vector of intermediate variables, and Y is the vector of output variables.

210

$$X = (x_1, x_2, ..., x_n)^T$$
(2)

$$U = \left(u_1, u_2, \dots, u_m\right)^T \tag{3}$$

$$Y = (y_1, y_2, ..., y_n)^T$$
(4)

where
$$y_i = f_i(x_1, x_2, ..., x_n, u_1, u_2, ..., u_m); \forall i = 1, ..., n$$
 (5)

$$u_{j} = \psi_{j}(x_{1}, x_{2}, ..., x_{n}); \forall j = 1, ..., m$$
(6)

Eq.s (2) to (6) represent non-linear relationships of a fuzzy multi-variable control model. In MS-FLC, the primary parameters of input variables are employed in the first stage, while in the second and the third stages both intermediate inputs from the first stage as well as external variables are utilized. Basically, the rules have multiple-inputs-single-output structure, where multiple inputs are used to produce a single output. Given these, the formation of rules in the three stages can be described as follows:

Stage 1
$$\begin{cases} R_{1} : If X_{1} is A_{1,x}^{1} \cap ... \cap X_{N}^{1} is A_{1,x}^{1} then Y_{1} is C_{1,y}^{1} \\ \\ R_{n_{1}} : If X_{1} is A_{n_{1},x}^{1} \cap ... \cap X_{N}^{1} is A_{n_{1},x}^{1} then Y_{1} is C_{n_{1},y}^{1} \end{cases}$$
 to the 2nd stage \Box (7)
Stage 2
$$\begin{cases} R_{1} : If Y_{1} is A_{1,x}^{2} \cap ... \cap X_{E}^{2} is A_{n_{1},x}^{2} then Y_{2} is C_{1,y}^{2} \\ \\ R_{n_{2}} : If Y_{1} is A_{n_{2},x}^{2} \cap ... \cap X_{E}^{2} is A_{n_{2},x}^{2} then Y_{2} is C_{n_{2},y}^{2} \end{cases}$$
 to the 3rd stage \Box (8)

Stage 3
$$\begin{cases} R_{1} : If Y_{2} is A_{1,x}^{3} \cap ... \cap X_{E}^{3} is A_{1,x}^{3} then Y_{3} is C_{1,y}^{3} \\ \\ R_{n_{3}} : If Y_{2} is A_{n_{3},x}^{3} \cap ... \cap X_{E}^{3} is A_{n_{3},x}^{3} then Y_{3} is C_{n_{3},y}^{3} \end{cases} defuzzification \qquad (9)$$

216 where:

- 217 $X_{(;)}, Y_{(;)}$: input and output variables respectively; n_1, n_2, n_3 : number of rules in stages 1, 2, 3 respectively
- 218 $A_{j,x}^{i}$: fuzzy number in antecedent part; i = 1, 2, 3: the stage; j : rule jth in each stage
- 219 x = 1,2,...,M: any fuzzy number in antecedent fuzzy sets; M is number of fuzzy sets in each input variable.
- 220 y = 1, 2, ..., O: any fuzzy number in conclusion fuzzy sets; O is number of fuzzy sets in each output variable.
- 221 N: number of input variables employed by 1st stage; $C_{j,y}^i$: fuzzy number in conclusion part.
- E: number of external input variables employed by stages 2 and 3

Note that in Eq.s (7) to (9) the rules are assumed homogeneous using the AND operator for simplicity. As will be seen in the following sections, in this MS-FLC the AND operator is predominant in the compositional operation, even though the OR operator is occasionally used.

The Eq.s (7), (8), and (9) are elaborated in section "Formation of Rules" below.

227 Formation of Rules

228 Stage 1: Evaluation of Current States of Traffic during Incidents

Stage 1 evaluates three principal quantities: congestion level, congestion mobility, and congestion status. The congestion mobility and congestion status specifically deal with the heavy congestion category, and the rule formation of these quantities are simple and straightforward. In the multiple input - single out (MISO) model, rules for the congestion level are characterized by two predicates (speed and density) in the antecedent, connected with an AND operator, and one predicate (congestion level) in the consequent. The general expression of rules is of the form:

If speed is $V_{(x)}$ AND density is $K_{(x)}$ then congestion level is $CL_{(x)}$. (10)

Fig. 2 shows an example of partition of the fuzzy sets for congestion level variable.



235

236

The collection of rules for congestion level is summarized in the rule decision matrix (Table 1). Some of combinations
such as "VeryHigh" speed - "VeryHigh" density, "VeryHigh" speed - "High" density, "High" speed - "VeryHigh" density,
... are unlikely to occur, thus they are removed from the Table.

240

241

Table 1. Rule decision matrix for congestion level (source: Toan and Wong, 2021)

(FF: Free flow, L: Light congestion, M: Moderate congestion, H: Heavy congestion, VH: Very heavy congestion)

		Density							
	Relation	VeryLow	Low	Medium	High	VeryHigh			
peed	VeryLow			Н	VH	VH			
	Low		М	М	Н	VH			
	Medium	L	L	М	Н	Н			
•1	High	FF	L	М	М				
	VeryHigh	FF	FF	L					

242

243 Stage 2: Prediction of Incident Traffic Conditions

In prediction of incident traffic conditions, it is essential to predict short-term traffic flow in the incoming period. This is an exogenous component from the MS-FLC (see Fig.1), but the prediction execution can be accomplished by a prediction software, and the result provided accordingly. As part of this research, Toan and Truong (2021) presented an efficient shortterm traffic flow prediction using support vector machine (SVM) and model training using nearest neighbor approach. The results are promising and proposals are made on extended research for online application.

Apart from the predicted traffic demand, the incident severity (the lane closure) is used to estimate the capacity remaining $\begin{pmatrix} C^* \end{pmatrix}$. Furthermore, the evaluation of the risk factor is necessary to anticipate the incident traffic conditions. The risk factor caters for external risks that exist exogenously with the prediction, ranging from the incident location, incident type, incident severity (capacity reduction), the time of day (peak/off-peak). The risk factor is decomposed into low/medium/high risk level. From the predicted traffic demand, the V/C^* is calculated, and then adjusted with the risk factor. There are 16 rules for this adjustment.

If predicted
$$\frac{V}{C^*}$$
 is Low and risk factor is high then the adjusted $\frac{V}{C^*}$ is medium (11)

The evolution of traffic trend depends heavily on the balance between traffic demand and supply, represented by the ratio of the predicted traffic demand (V) upstream and the capacity remaining (C^*) at the incident location. Fig. 3 shows the membership functions for adjusted $\frac{V}{C^*}$ ratio.



258

259

Fig. 3. Membership functions for adjusted $\frac{V}{C^*}$ ratio

Given the congestion level estimated in the first stage and the adjusted $\frac{V}{C^*}$ ratio, the MS-FLC evaluates the predicted congestion level. An example of rule of this type is as follows:

If adjusted $\frac{V}{C^*}$ is High and CongestionLevel is Light then predicted-CongetionLevel is Moderate. (12)

The collection of predicted congestion level consists of 16 rules. Note that in this sub-stage, the variable CongestionLevel indicates the prevailing current congestion level, which does not include the Heavy congestion level since it is tracked directly from the 1st stage into the 3rd stage of the MS-FLC.

265 Stage 3: Recommendation of control action

266 Stage 3 receives the evaluated and predicted traffic conditions from previous stages, and other traffic and incident information 267 to provide recommended solutions. The expressway operation management during incidents undertakes important tasks, including the dissemination of prevailing information to motorists, the regulation of ramp access, the control of route 268 269 diversion, and the management of queues. The tasks employ appropriate control measures to target the control goals: the 270 amelioration of the mainline congestion and prevention of excessive ramp queues. The goals are translated into specific 271 measurable and tangible objectives such as to maximize mainline utilization, to prevent mainline congestion, to prevent 272 excessive ramp queue, or to balance between objectives. Subsequently, the objectives are evaluated using specific measures 273 of effectiveness (MOEs) as described in Section on "Results and Analysis". Since the two objectives may be conflicting to 274 each other, rules should be designed to compromise them at a balance point. For incident management, the control objectives 275 target efficient incident responses for the mainline without incurring excessive ramp queues.

Table 2 summarizes the decision rules for the local ramp control strategy. Each rule is a mapping between two (three) predicates in the rule conditions and one predicate in the rule conclusion. The rule conditions are joined with AND connectives. The rule conclusion reflects the control action that infers ramp flow based upon the rule conditions in the direction of the key control objective that elaborates the control goals: in correspondence to the key control objective, the conditions of the rules consider the traffic condition (congestion level, CL) upstream of the incident (downstream of the

ramp), the traffic demand (indicated by the V/C^* ratio) upstream of the ramp, and the ramp queue (see Fig. 5 later). For 281 scenarios such that the traffic condition upstream of the incident and the $\frac{V}{C^*}$ upstream of the ramp favor high ramp flows, 282 283 the rules can be generated regardless of the queue status. Specifically, if the traffic condition upstream of the incident is Free-284 flow or Light and traffic demand is Low/Medium, the ramp flow is set to High/Very_high level so as to maximize mainline *utilization* (rules 1, 2, 7). In contrast, if the traffic demand (V_{C^*} ratio) upstream is *High/Very_high* the ramp flow is set to 285 286 Low/Very_low levels to prevent mainline congestion (rules 3, 6, 9, 10, 18, 20, 23, 24). In addition, the ramp flow is adjusted 287 according to the ramp queue status so as to maintain acceptable ramp queue (rule 4), to prevent excessive ramp queue (rules 288 5, 11, 21, 22), or to maintain a balance between objectives (rules 8, 13, 14, 19). Finally, if the traffic on the mainline is 289 congested, the restriction of the ramp flow is to target preventing a secondary ramp queue at the ramp merge (rules 12, 15, 290 16, 17). The reason for this restriction is that when the mainline is congested, the ramp traffic will hardly find an acceptable 291 gap to join the mainline, so a secondary queue of the metered vehicles may form spontaneously. If a secondary queue persists, 292 ramp metering is not beneficial. At the extreme, vehicles in the secondary queue may try to encroach the mainline, breaking 293 down traffic upstream of the ramp and creating safety risk. Therefore, in the presence of a secondary queue, it is imperative 294 that the vehicles be stored on the ramp to wait for an opportunity in the next period rather than being metered. The inputs are 295 combined in such a way that predicates are scaled gradually over the input domains, and the outputs are translated elegantly 296 from one fuzzy value to another. For example, in rules 7, 8, and 9, given the *Light* congestion level, when the V_{C^*} changes 297 from Low to Medium to High, the Ramp_Flow changes from High to Medium to Low, respectively.

298

299

Table 2. Decision table for rules with local ramp control

(Note: SQ-HC: short queue-heavy congestion)

	R	lule condition		Rule conclusion	
Rule	Congestion level (CL) upstr. of the incident	$\frac{V}{C^*}$ upstr. of the ramp	Ramp Queue	Ramp Flow	Key Control objective
1	Free-flow	Low		Very_high	Maximize mainline utilization
2	Free-flow	Medium		High	Maximize mainline utilization
3	Free-flow	High	Short	Low	Prevent mainline congestion
4	Free-flow	High	Medium	Medium	Maintain acceptable ramp queue

5	Free-flow	High	Long	High	Prevent excessive ramp queue
6	Free-flow	Very_high		Low	Prevent mainline congestion
7	Light	Low		High	Maximize mainline utilization
8	Light	Medium		Medium	Balance between objectives
9	Light	High	Short	Low	Prevent mainline congestion
10	Light	High	Medium	Medium	Prevent mainline congestion
11	Light	High	Long	Medium	Prevent excessive ramp queue
12	Light	Very_high		Very_low	Prevent secondary queue
13	Moderate	Low		Medium	Balance between objectives
14	Moderate	Medium		Medium	Balance between objectives
15	Moderate	High	Short	Low	Prevent secondary queue
16	Moderate	High	Medium	Low	Prevent secondary queue
17	Moderate	High	Long	Medium	Prevent secondary queue
18	Moderate				
	Widderate	Very_high		Very_low	Prevent mainline congestion
19	SQ-HC	Very_high Low		Very_low Medium	Prevent mainline congestion Balance between objectives
19 20	SQ-HC SQ-HC	Very_high Low Medium	Short	Very_low Medium Low	Prevent mainline congestion Balance between objectives Prevent mainline congestion
19 20 21	SQ-HC SQ-HC SQ-HC	Very_high Low Medium Medium	 Short Medium	Very_low Medium Low Medium	Prevent mainline congestion Balance between objectives Prevent mainline congestion Prevent excessive ramp queue
19 20 21 22	SQ-HC SQ-HC SQ-HC SQ-HC	Very_high Low Medium Medium Medium	 Short Medium Long	Very_low Medium Low Medium Medium	 Prevent mainline congestion Balance between objectives Prevent mainline congestion Prevent excessive ramp queue Prevent excessive ramp queue
 19 20 21 22 23 	Noderate SQ-HC SQ-HC SQ-HC SQ-HC	Very_high Low Medium Medium Medium High	Short Medium Long	Very_low Medium Low Medium Medium Low	 Prevent mainline congestion Balance between objectives Prevent mainline congestion Prevent excessive ramp queue Prevent mainline congestion
 19 20 21 22 23 24 	SQ-HC SQ-HC SQ-HC SQ-HC SQ-HC SQ-HC	Very_high Low Medium Medium Medium High Very_high	 Short Medium Long 	Very_low Medium Medium Medium Low Very_low	Prevent mainline congestionBalance between objectivesPrevent mainline congestionPrevent excessive ramp queuePrevent excessive ramp queuePrevent mainline congestionPrevent mainline congestion

300 Development of the TSC

This section presents the development and validation of a Traffic Simulator and Control (TSC) model and the implementation and evaluation of the MS-FLC framework presented in previous sections. The TSC model (Fig. 4) is developed in SIMULINK in MATLAB, following the decision-making logic for incident-related traffic control 304 presented in the conceptual model (Fig. 1). The TSC consists of two main components (Fig. 4): the car-following

305 model (CFM), and the traffic controller (TC).



306

307

Fig. 4. Conceptual model of the TSC

The CFM simulates the car-following behavior and delivers the aggregated traffic parameters to the TC for traffic control. In this study, the CFM is developed using the modelling concepts provided by Gazis-Herman-Rothery (GHR) type of models. Although the CFM simulation keeps track of individual vehicles, only aggregated traffic variables (flow rate, density, total travel time, mean speeds - see the MOEs in Tables 4-7 also) are parameters of interest. In other words, by using the microscopic simulation, the model explains the macroscopic behavior of systems and obtains macroscopic traffic metrics.

Although individual vehicles are tracked, the TSC functions more like a macroscopic traffic simulation model since only aggregated traffic variables, which are the parameters of interest, are generated. By including the CFM component in the model, the dynamic longitudinal interactions between vehicles, namely car-following behaviors, are replicated. The TC receives the aggregated outputs for traffic control purposes. The traffic on the multi-lane expressway where the data was collected is represented as an equivalent single-lane system for model calibration and validation.

In a multi-lane highway, a standard microscopic traffic simulation package examines both car-following and lanechanging behavior. Unfortunately, SIMULINK lacks the ability to capture lane-changing behaviors. Lane-changing maneuvers may have a significant impact on the speeds and travel times of vehicles in the traffic stream in free-flow conditions, but there are few lane-changing opportunities in congested conditions. Furthermore, because the parameters of interest are the overall macroscopic traffic variables that are averaged across lanes, they may not be highly sensitive to cars changing lanes, and traffic control for non-recurring congestion often concentrates on congested scenarios. As a result, through the calibration of its parameters, the CFM developed in this research implicitly integrates lane-changing effects.

- 328
- 329



330

Fig. 5. The MS-FLC in SIMULINK

An iterative process of calibration simultaneously refines the model's parameters, ensuring that the model 331 332 accurately replicates real-world behavior. The calibration of the CFM identifies the most influential parameters: desired gap, gain factor for acceleration, gain factor for deceleration, maximum acceleration, maximum deceleration, 333 334 speed limit, and reaction time. Having calibrated, the CFM validation was performed at the macroscopic levels where speeds and flow rates for simulated platoons are aggregated in one-minute intervals and are compared with those of 335 336 field data on a segment of the Singapore's Pan Island Expressway (PIE) under various traffic conditions (free-flow, medium congestion and heavy congestion). The result shows that the simulated speed is not significantly different 337 338 from the field speed (at the significance level) for both upstream and downstream segments, and the aggregated flow 339 rate discrepancies fall within small ranges.

The designed MS-FLC (Fig. 5) was embedded in the TC component for MS-FLC evaluation. Over different traffic situations and incident scenarios, the MS-FLC performance was compared to that of the No-control scenario and the ALINEA ramp controller. For the MS-FLC to execute in its totality, the model requires predicted short-term traffic flow for the incoming period to anticipate incident related traffic condition. The data are provided by an external SVM short-term traffic flow prediction component. The SVM is linked with a real-time database so that data can be continually retrieved for the MS-FLC operation using the rolling-horizon approach proposed by Peeta and Mahmassani (1995). As stated earlier, although the SVM prediction performance is promising, additional effort

347 need to be devoted to applying the SVM model for online application. Thus, for the time being, in the model evaluation

348 section below, the MS-FLC use the data in the current interval to project the future state. In this experiment ALINEA

349 (ALINEA\Q) control algorithm is used to compare with MS-FLC, thus the control algorithms must have the same

350 simulation and network setting as described below.

351 Model Evaluation

352 General Settings

353 It would be preferable to use observed data with a real network to explore the model behavior under various conditions for model evaluation. However, obtaining data from actual sites is technically complex, time consuming, and very costly. 354 355 Simulated traffic, on the other hand, may be duplicated from one run to the next, making comparisons between scenarios 356 simple. The use of a generic network for simulation-based evaluation is a viable option that allows for more flexibility in examining various traffic conditions and incident scenarios, while the criteria for evaluating the success of control algorithms 357 358 can be simply and uniquely obtained. In this regard, the FLC control algorithm evaluated in this part uses a simulated study segment as shown in Fig. 6. The study segment is modelled after the validated site (section 80007774) that was previously 359 360 described. The segment comprises three links: one upstream of the ramp, one downstream of the ramp, and one upstream of 361 the incident (downstream of the ramp). The majority of measurements for local ramp control are collected in the vicinity of 362 the incident, notably the upstream and downstream links. The lengths of the links used in this experiment are L_{upstr} =1,000m, $L_{downstr} = 500m$. The expressway's capacity is reduced as a result of the lane-blocking incident, and local ramp control is 363 implemented to regulate traffic demand from the ramp in order to avoid or alleviate mainline congestion. 364



365

366

Fig. 6. Layout of the study segment

The inputs in evaluation involve two pairs of time-dependent O_1D_1 demands, speed profile of the first vehicles, and timevarying splits at the diversion route. The time-varying splits are specifically considered in the rules in the FLC algorithm. The evaluation investigates a wide range of traffic conditions and incident situations. The traffic O_1D_1 flows are loaded at Low, Medium, and High demand levels, the values of which are defined based on local conditions. In addition to traffic 371 conditions on the expressway and on the ramp, the evaluation investigates various incident scenarios, including capacity
 372 reduction and incident location.

In this experiment the ramp is assumed to have a storage capacity of 60 vehicles. Once the ramp queue reaches this level, the urban traffic will not join the ramp queue but will be diverted to the surface streets and enter the expressway through downstream ramps. The availability of diversion alternatives encourages the local traffic to utilize the parallel urban streets in case of critical mainline traffic conditions.

The parameters of interest used for control and evaluation are aggregated variables including traffic flow rate $q_{(j)}(t)$, speed $v_{(j)}(t)$, and density $k_{(j)}(t)$ for every interval *t*, where the (;) denotes the locations upstream and downstream of the incident. Apart from that, the queues on expressway and on the ramp are also collected. The total study time is about 90 minutes, including: the first one third part is normal traffic, the second one third part is incident period, and the last one third part is normal traffic again. There are several MOEs that can be used as the evaluation criteria, including total travel time on expressway, total waiting time on the ramp, total time spent in the system, total travel distance, average speed on expressway, and mean density.

384 The basic parameters of the simulation: simulation time: 90 minutes, including:

- From the 1st min. to 30th min.: normal traffic
- From the 31st min. to 60th min.: incident period
- From the 61st min. to 90th min.: normal traffic
- Evaluation interval: every 10 seconds
- Evaluation period: from the 16th min. to 90th min.

390 To achieve a high level of representation and accuracy, the vehicle's acceleration, speed and position are updated every391 0.1 second.

392 Three control methods can be considered: No control; $ALINEA \setminus Q$ control, and FLC control. ALINEA is the most widely 393 used technique in the close-loop control (Papageorgiou et al. 1991). ALINEA determines the metering rates such that the 394 traffic state on the expressway approaches a pre-defined condition. Developed as an enhancement of ALINEA, the ALINEA Q 395 (Smaragdis and Papageorgiou 2003) incorporates ramp control with ramp queue management by considering two metering 396 rates. The first rate is calculated exactly the same as that in the ALINEA algorithm, while the second rate is calculated so as 397 to maintain the ramp queue within a desirable queue length. The FLC control monitors the ramp flow by considering both 398 the congestion level of the expressway and the ramp queues, with priority given to the mainline traffic. Results from initial 399 scenarios will be used to train the FLC before the actual evaluation.

400 Since ALINEA is considered an efficient local-ramp control algorithm for monitoring the mainline traffic, in this 401 experiment ALINEA is used to compare with FLC. ALINEA uses the measured occupancy at a loop detector downstream of 402 the ramp, and regulates the ramp flow based on the difference between the measured occupancy and the optimal set point

403 occupancy (Papageorgiou et al. 1991). The Eq. used to calculate the metering rate for time interval t is:

$$q_r(t) = q_r(t-1) + K_R [O_{opt} - O_{down}(t-1)]$$
(13)

404 where:

405 $q_r(t)$ and $q_r(t-1)$: metering rates of the current and previous intervals, respectively.

406 K_R : regulator parameter. Field experiment has shown that ALINEA has not been very sensitive to the choice of K_R , and

407 the typical value of K_R is 70 veh/h (Papageorgiou et al. 1991).

408 *O*_{opt}: set point optimal occupancy, which is set to obtain optimal operation (Papageorgiou et al 1991).

409 $O_{down}(t-1)$: occupancy downstream in the previous interval.

Since the standard ALINEA algorithm targets the optimal occupancy at the immediate detector downstream of the ramp, it uses the point measurement. Therefore, in this experiment the average occupancy for the whole section from the incident location to the ramp is recommended to capture the spatial effect of the incident. The average occupancy is estimated from the average density.

$$O_{down}(t) = (L+d) \times k(t) \tag{14}$$

414 where *L* is the average vehicle length, *d* is the length of the detector. The average density k(t) in each evaluation interval 415 is calculated by the ratio between the number of vehicles on the link and the length of the segment.

The average vehicle length is the arithmetic mean of lengths of various vehicle types, which can be derived from the vehicle composition. Eq. (14) holds true when the vehicles have constant speeds. In congested condition this assumption is not valid, and Eq. (15) will be used instead:

$$O_{down}(t) = \frac{\sum_{i} (L_i + d) / v_i}{T}$$
(15)

419 where L_i is the length of vehicle type *i*; v_i is the vehicle speed; *T* is the period of measurement.

420 The critical occupancy O_{cr} is the occupancy associating with the maximum flow rate. It was determined from an 421 empirical volume-occupancy relationship, established from 227 simulated records, and the resulting $O_{cr} = 26\%$ was 422 obtained. O_{opt} is taken as 24%, slightly lower than O_{cr} .

423 The traffic controller of the ALINEA algorithm is designed in SIMULINK and is shown in Fig. 7.



424	
425	

Fig. 7. ALINEA controller in SIMULINK

426	In the evaluation, the setting of traffic demand is evaluated approximately based on the $V/_{C^*}$ ratio, where V is the traffic
427	volume, and C^* is the remaining capacity. Although technically the traffic under various situations can be investigated, for
428	purposes of discussion in this paper, only a high-expressway demand scenario, wherein the traffic demand is about 1,000-
429	1,100 veh/h/lane, is presented here. This scenario encompasses several cases in which the high level of mainline traffic
430	demand is associated with various levels of ramp traffic demand, capacity reduction, and incident location. More specifically,
431	the following cases are investigated:
432	• Case 1: Medium ramp demand; and
433	• Case 2: High ramp demand.
434	Since the experiment focuses on congested conditions, Case 2 was extended to:
435	• Case 3 with more severe capacity reduction (less remaining capacity); and
436	To see the effect of the incident location, Case 3 was extended to:
437	• Case 4 with the incident location moved upstream, to 500m downstream of the ramp.
438	The settings in each case are listed in Table 3.

439

Table 3. Settings in each case

	Mainline traffic demand (veh/h/lane)	Ramp demand (veh/h)	Remaining capacity <i>C</i> [*] (%)	Incident location (distance at downstream of the ramp)
Case 1	1,000-1,100	300±10%	45-50%	1,000m
Case 2	1,000-1,100	400±10%	45-50%	1,000m
Case 3	1,000-1,100	400±10%	30-40%	1,000m
Case 4	1,000-1,100	400±10%	30-40%	500 m

440 Measures of effectiveness (MOEs)

441 The TSC uses the following measures of effectiveness as the evaluation criteria:

442 <u>a) Total travel time on the expressway, TTT (veh.h)</u>

443 The TTT is the sum of travel times of individual vehicles. In SIMULINK, the TTT is the sum of the number of vehicles in 444 the expressway N(t) over time in successive intervals:

$$TTT = \sum_{t=t_0}^{t=T} N(t) \times \Delta_t$$
(16)

445 The TTT is a principal evaluation criterion. The calculation of the TTT allows the comparison of the total time spent in the 446 system. The lower the TTT indicates the positive signal, providing that higher throughput and higher speed are also obtained. 447 Nevertheless, if the lower TTT is the result of too restrictive a control method that produces a lower throughput, this "saving"

is misinterpreted. Therefore, the TTT should be evaluated in accordance with the other MOEs.

449 b) Total waiting time on the ramp, TWT (veh.h)

450 The TWT is the accumulated waiting time of vehicles in the ramp queue due to the control regulation. Like TTT, TWT is the

451 sum of the number of vehicles in ramp queue Q_r over time in successive intervals:

$$TWT = \sum_{t=t_0}^{t=T} Q_r(t) \times \Delta_t$$
(17)

Unlike TTT, TWT is a secondary criterion an incident management strategy normally sets a higher priority for the expresswaythan the ramp traffic.

454 <u>c) Total time spent in the system, TTS (veh.h)</u>

455 The TTS is the total time all vehicles spend in the system during the simulation period, being the sum of the TTT and TWT.

$$TTS = TTT + TWT$$
(18)

456 d) Total travel distance, TTD (veh.km)

- 457 The TTD is the sum of distances travelled by individual vehicles during the simulation. In SIMULINK, the TTD is calculated
- 458 as the sum of the total of travel distances upstream and downstream sections in successive intervals.

$$TTD = \sum_{t=t_0}^{t=T} \left[N_{up}(t) \times \overline{V}_{up}(t) + N_{down}(t) \times \overline{V}_{down}(t) \right] \times \Delta_t$$
⁽¹⁹⁾

459 where $N_{up}(t)$ and $N_{down}(t)$ denote the number of vehicles in upstream and downstream links during interval t; $\overline{V}_{up}(t)$ 460 and $\overline{V}_{down}(t)$ are the space mean speeds during the same period.

- 461 Like TTT, TTD is a primary MOE since it indicates the level of "productivity" the expressway yields. It encompasses both
- 462 the mainline throughput and average speed.

463 e) Average speed on expressway, MS (km/h)

464 The MS on the expressway is among the most important criteria since it represents the dynamics of a vehicle's motion. The

465 average speed is calculated as the ratio of TTD and TTT.

$$MS = \frac{TTD}{TTT}$$
(20)

466 where TTD and TTT are associated with the same number of vehicles (see Block 2, Appendix D).

467 <u>f) Mean density, MD (veh/km)</u>

468 Like speed, MD is a primary indicator of congestion level. The mean density is the arithmetic mean of traffic densities k(t)469 in the network in successive intervals.

$$MD = \frac{\sum_{t=0}^{n} k(t)}{N}$$
(21)

470 where *N* is the number of simulated intervals. Since the density is determined for upstream and downstream segments 471 separately, the traffic density k(t) in the network in an interval *t* is calculated as the weighted mean of densities on upstream 472 and downstream segments:

$$k(t) = \frac{L_{upstr} \times k_{upstr}(t) + L_{downstr} \times k_{downstr}(t)}{L_{upstr} + L_{downstr}}$$
(22)

where L_{upstr} and $L_{downstr}$ are the lengths of upstream and downstream segments, respectively. In Section 8.4.3 the two segments respectively have the lengths of 1,000 m and 500 m, excepting for the Scenario "*High demand, Case 4*" where the incident location is assumed to move upstream, the length of the segments change, i.e. $L_{upstr} = 500$ m and $L_{downstr} = 1,000$ m.

477 Apart from the described measures, the simulation considers the maximum length of queues on the expressway Q_{exp} and on 478 the ramp Q_r .

The TSC model was developed in SIMULINK in MATLAB. SIMULINK is a graphical programming language that offers modelling, simulation and analysing of dynamic systems under a Graphical User Interface (GUI) environment. SIMULINK facilitates easy communication between the simulation with external applications. In SIMULINK the CFM and the TC are harmonized and integrated in a close-loop control system, with the control effects (TC outputs) fed-back as inputs to the and CFM for real-time applications. Embedded in SIMULINK the simulation parameters can be easily specified and altered for various scenarios and sensitivity analysis.

485 Results and Analysis

Tables 4 to 7 show the values and percentile changes of the MOEs. For temporal MOEs, including total travel time (TTT), total waiting time (TWT), total time spent (TTS), a negative sign of percentile change indicates time saving. For spatial MOEs, including mean density (MD), maximum length of queues on the expressway (max Q_exp), and maximum length of queues on the ramp (max Q_ramp), a negative sign indicates improvement. For the remaining attributes, including total travel distance (TTD), and average speed (MS), a positive sign is a positive indication of the related parameter.

491 Case 1: Medium Ramp Demand

Table 4 lists the results from Case 1. The table shows that in general under both *ALINEA* and *FLC* significant benefits were achieved. *ALINEA* gained a *TTT* saving of 13.13%, an increase in *MS* of 15.12%, and a reduction in *MD* of 13.12%, compared to *No control*. The algorithm also enjoyed a substantial reduction in *max Q_exp* of 32.28%. Nevertheless, *ALINEA* suffered considerable long *TWT* of 9.54 veh.h, and an excessive ramp queue (*max Q_ramp*) of approximately 46 vehicles.

496

		No Control	rol ALINEA		FLC		ALINEA\Q vs FLC
MOE	Unit						
		value	value	% change	value	% change	% change
TTT	veh.h	55.62	48.32	-13.13	48.36	-13.06	-0.08
TWT	veh.h	0	9.54		6.56		31.24
TTS	veh.h	55.62	57.86	4.03	54.91	-1.28	5.10
TTD	veh.km	2541.43	2541.43	0	2541.43	0	0.00
MS	km/h	45.69	52.6	15.12	52.56	15.02	0.08
MD	veh/km	29.55	25.67	-13.12	25.69	-13.05	-0.08
max Q_exp	veh	112.47	76.05	-32.28	77.62	-30.99	-2.06
max Q_ramp	veh	0	46.54		20.55		55.84

Table 4. MOEs for Case 1

497

The *FLC* obtained a compatible level of benefits: the improvements in the *TTT*, *MS*, and *MD* were 13.06%, 15.02%, and 13.05%, respectively. As compared to *ALINEA*, the *TWT* and *max* Q_ramp under *FLC* were less severe, which leads to a saving in *TTS* of 1.28% compared to a loss of 4.03% under *ALINEA*. The *TTDs* were the same since the traffic states were similar across three control methods at the beginning and at the end of the evaluation period (there was no queue on the mainline and on the ramp at these time points).

503 Case 2: High Ramp Demand

504 To explore how the control algorithm work under critical conditions, the experiment was carried out with high demands on 505 both expressway and ramp in Case 2. The results from Case 1 (Table 2) show that the standard ALINEA gained substantial 506 benefits to the mainline, where the key MOEs such as TTT, MS, MD and max Q_exp were improved considerably. To some 507 extent, ALINEA even slightly outperformed FLC control with respect to the mainline conditions. Nevertheless, the ALINEA algorithm shows that the method merely targets benefits for the mainline without considering the status of the ramp traffic. 508 Under heavy ramp demands, the mechanism used in the standard ALINEA would likely induce intolerable traffic conditions 509 510 on the ramp. In practice, the principle of traffic control should be such that smooth expressway travel can be achieved, while maintaining a reasonable ramp traffic status. In incident management in particular, the control objectives should target 511 512 efficient incident responses to the mainline traffic without incurring excessive ramp queue length. Therefore, the ALINEA\Q is used in Case 2 instead. 513

Table 5 summarizes the results of the simulation for Case 2. The table shows that both *ALINEA**Q* and *FLC* control methods achieved considerable improvements: *ALINEA**Q* gained a *TTT* saving of 13.92%, an increase in the *MS* of 15.61%, and a decrease in the *MD* of 13.50%. In particular, *ALINEA**Q* handled the ramp queue better than *FLC* and slightly better than the standard *ALINEA* under Case 1 (Table 4).

518

		No Control	ALINEA\Q		FLC		ALINEA\Q vs FLC
MOE	Unit						
		value	value	% change	value	% change	% change
TTT	veh h	70.31	60.52	_13.02	55 14	-21.58	8 80
111	ven.n	70.51	00.52	-13.92	55.14	-21.56	0.07
TWT	veh.h	12.91	22.62	75.24	24.71	91.4	-9.24
TTS	veh.h	83.22	83.15	-0.09	79.85	-4.05	3.97
TTD	veh.km	2728.41	2715.12	-0.49	2719.53	-0.33	-0.16
MS	km/h	38.8	44.86	15.61	49.32	27.1	-9.94
MD	veh/km	37.42	32.36	-13.5	29.44	-16.7	9.02
max Q_exp	veh	153.87	135.22	-12.12	92.73	-39.73	31.42
max Q_ramp	veh	33.4	45	34.73	50	49.7	-11.11

Table	5.	MOEs	for	Case	2
-------	----	------	-----	------	---

The *FLC* control alternative, with an exception of the ramp-related attributes, gained higher benefits than *ALINEA\Q*.
The improvements in *TTT, MS,* and *MD* were 21.58%, 27.1%, and 16.7%, respectively. In particular, *FLC* also gained a
reduction in the *TTS* of 4.05%.

523 Results from Case 1 and Case 2 show that there exist excessive long queues on the mainline. In Table 5 in particular, the 524 expressway queues under No control, ALINEAQ, and FLC were 153.87, 135.22, and 92.73 vehicles, respectively. This is 525 partially attributed to the implicit assumption that the ramp closes only when the mainline queue reaches the ramp. If the 526 incident occurs far from the ramp, this passive type of ramp closure will tolerate a very severe mainline condition. It should 527 be noted that if a long queue exists on the mainline, additional discharge from the ramp may not benefit the ramp traffic but 528 aggravate the mainline conditions, thus a longer time will be required for the mainline traffic to dissipate. To minimize 529 extreme congestion, an active action of ramp closure should be conducted from a control standpoint. Therefore, in Case 3 530 under ALINEA Q and FLC the ramp closure is set when the mainline queue reaches 50% of the length of the upstreamincident segment, while this feature of operation is not available under No control. 531

532 Table 6 lists the results from Case 3. The incident is assumed to create a more severe capacity reduction (remaining capacity within 30-40%). The table shows that benefits of ALINEA Q and FLC obtained for the mainline in this Case were, 533 534 in general, higher than the previous Cases. Compared to No control, ALINEA\Q gained a TTT saving of 22.14%, an increase in the MS of 26.82%, a reduction in the MD of 23.44%, and a cut down in the max Q exp of 41.86%. The FLC benefits were 535 536 even more profound with improvements in TTT, MS, MD, and max Q_exp of 23.13%, 27.98%, 23.11%, and 42.61%, 537 respectively. The improvements of ALINEAQ and FLC were certainly due to a strong regulation of the ramp traffic with active response to the mainline conditions. The results under No control also indicate that without strong control intervention, 538 the system performances may deteriorate seriously. Despite that, with the early ramp closure subjected to the mainline queue, 539 540 it is certain that ALINEA\Q and FLC impose more TWT, and more vehicles have to be diverted from entering the ramp.

541

Table 6. MOEs for Case 3

MOE	TT '	No Control	ALINI	EA\Q	FL	С	ALINEA\Q vs FLC
MOE	Unit	value	value	% change	value	% change	% change
TTT	veh.h	71	55.28	-22.14	54.57	-23.13	1.28
TWT	veh.h	20.28	25.75	26.99	23.9	17.86	7.18
TTS	veh.h	91.28	81.04	-11.22	78.47	-14.03	3.17
TTD	veh.km	2543.77	2511.95	-1.25	2502.36	-1.63	0.38
MS	km/h	35.83	45.44	26.82	45.85	27.98	-0.90
MD	veh/km	37.89	29.01	-23.44	29.14	-23.11	-0.45
max Q_exp	veh	181.85	105.72	-41.86	104.37	-42.61	1.28
max Q_ramp	veh	60	60	0	60	0	0.00

542 Case 4: Incidence Location Changed

543 Case 4 is associated with the mainline demand in the range of 1,000-1,100 veh/h/lane, the ramp demand in the range of 400 \pm 10% veh/h, and the remaining capacity C^* between 30-40%. The incident occurred at 500 m downstream of the ramp, which 544 is closer than those in Cases 1 to 3. Table 7 summarizes the results from the simulation, which shows that the benefits from 545 546 both ALINEA Q and FLC were less profound than the previous Cases: ALINEA Q gained a TTT saving of 6.49%, an increase 547 in the MS of 5.6%, a reduction in the MD of 6.72%, and a reduction in the max Q_exp of 13.97%. The improvements in TTT, MS, MD, and max_q_exp under FLC were 11.34%, 11.88%, 13.13%, and 19.69%, respectively, that are remarkably higher 548 549 than ALINEA\Q. Nevertheless, ALINEA\Q and FLC incurred 22.27% and 10.19% more of TWT than No control, respectively. 550 In particular, the two control algorithms yielded 1.25% and 0.81% of the total mileage TTD less than No control. This is 551 probably due to the fact that the when the ramp queue reaches the ramp's physical storage capacity, vehicles that arrive at the 552 ramp will not proceed to join the queue, but be diverted to the parallel street.

553

Table 7. MOEs for Case 4

MOF	Unit	No Control	ALINEA\Q		FLC		ALINEA\Q vs FLC
MOL		value	value	% change	value	% change	% change
TTT	veh.h	57.41	53.68	-6.49	50.89	-11.34	5.20
TWT	veh.h	23.05	28.18	22.27	25.4	10.19	9.87
TTS	veh.h	80.45	81.86	1.75	76.29	-5.18	6.80
TTD	veh.km	2509.66	2478.26	-1.25	2489.29	-0.81	-0.45
MS	km/h	43.72	46.16	5.6	48.91	11.88	-5.96
MD	veh/km	30.72	28.65	-6.72	26.68	-13.13	6.88
max Q_exp	veh	127.19	109.42	-13.97	102.14	-19.69	6.65
max Q_ramp	veh	60	60	0	60	0	0.00

Through the evaluation in comparison with the *No-control* scenario and *ALINEA* (*ALINEA*\Q) ramp control algorithm, 554 555 it can be concluded that the proposed MS-FLC with the FLC controller showed substantial benefits. Particularly, under high traffic demand and severe capacity reduction, the FLC brings higher travel time savings as well as improvements of traffic 556 557 conditions on both the mainline and ramp. Not only does the FLC outperform ALINEA\Q in managing ramp traffic, it also 558 outperforms ALINEA\Q in managing the mainline flow under critical incident congestion. However, it is noted that the 559 benefits of control interventions (ALINEA and FLC) depend on the magnitude of traffic demand and incident situation. In 560 general, under high traffic demand and critical incident conditions, more significant gains can be realized than under favorable 561 conditions. This comparison is likewise based on a simplified segment with a one-lane ramp. The assumption that the lane

has a storage capacity of 60 vehicles should be modified accordingly, and the benefits (savings in travel times, distances, and
 so on) should be adjusted accordingly.

564 Sensitivity Analysis

The findings of the simulation experiment in varied traffic demand (low, medium, high) and incident scenarios are presented in the previous section (capacity reduction, incident location). Nonetheless, the scenarios were coupled with predetermined hypothetical network designs (a 1.5-kilometer network length (upstream section = 1.0 km, downstream section = 0.5 km) and a 60-vehicle ramp storage capacity), and a 90-minute simulation time. These network and simulation settings have a substantial impact on model performance, and it is unclear whether the control methods' comparative performance will remain valid if the input parameters change.

A sensitivity analysis is conducted to explore the effects of changes in these parameters on the comparative performance of the control approaches and to enhance confidence in the models' performance in an uncertain environment. Because these parameters are unrelated, the sensitivity analysis is carried out separately for each one, so that one parameter is altered while the others remain constant in each run.

575 The simulation parameters are changed as follow:

a) Network length: The simulated mainline consists of the upstream and downstream sections of the incident. Since
the impacts of the incident can mostly be observed upstream, this analysis investigates how the change with a change
in the length of the upstream section. Four scenarios are extended to the length of the upstream section increased
from 1.0 km to 1.5, 2.0, 2.5, and 3.0 km, respectively. The length of the downstream section in the four scenarios
remains at 0.5 km.

b) Ramp storage capacity: the ramp storage capacity of 60 vehicles in the simulation is now changed to 20, 40, 80 and
100 vehicles, respectively.

c) Simulation time: The previous simulation investigated the model performances for the simulation time of 30 minutes
 for each of the pre-incident, incident, and post-incident periods (named hereafter as scenario "30-30-30"). To explore
 how the improvement in the mean speed changes with simulation time, the simulation time is extended to two
 scenarios 30-60-30 and 30-60-60 minutes, respectively.

587 Since the use of all MOEs in this analysis would be very confusing, the mean traffic speed could be the best MOE in this 588 sensitivity analysis given that the mean speed is a key parameter that reflects the operational condition on the mainline. The 589 relative change in the mean speed of the control methods over "No control" is used and is calculated as:

$$\Delta_{\rm MS}^{\rm i} = 100 \times \frac{MS_i - MS_{No}}{MS_{No}} (\%) \tag{16}$$

590 where *i* denotes either *ALINEA**Q* or *FLC* method, MS_i denotes the mean speed under the control method *i*, and MS_{No} 591 denotes the mean speed under "No control".

592 The sensitivity analysis is performed for the Case 3 "High expressway and ramp demands, severe capacity reduction 593 (C*=30-40%). Table 8 show the Δ_{MS} versus the length of upstream section. The Table indicates that both ALINEA\Q and 594 FLC are highly sensitive to the length of the upstream section, and the superiorities of the control methods over No control 595 deteriorate as the network length increases. For a relatively short simulated network (the length of the upstream section = 596 1.0-1.5 km), a small change in the network length may lead to a large change in Δ_{MS} , but for a relatively long simulated 597 network (the length of the upstream section = 2.5-3.0 km) the change in Δ_{MS} against a change in the network length is smaller. A possible reason under this phenomenon could be due to the fact that for a given traffic demand and incident 598 599 parameters, when the upstream section is shorter, the traffic condition is more critical. By contrast, when the network length 600 is large the traffic condition is less severe, and the effectiveness of the control is lower.



Table 8. $\Delta_{\rm MS}$ versus the length of upstream section

Control method	Length of upstream section (km)						
Control method	1.00	1.50	2.00	2.50	3.00		
ALINEA\Q	26.82	14.62	10.84	8.84	7.42		
MS-FLC	27.98	20.84	16.68	14.16	12.45		

Table 9 shows that in both control methods the Δ_{MS} varies slightly in the range 23-29%, and the values of Δ_{MS} increase as the ramp storage capacity increases. A possible reason could be that when the ramp storage capacity increases, the ramp can accommodate more vehicles, hence fewer vehicles have to divert from the ramp. Consequently, given a long ramp and regardless of the control method, more vehicles can be metered into the mainline. In both cases, it is obvious that the MS-FLC consistently outperforms the ALINEA\Q control algorithm.

607

Table 9. Δ_{MS} versus the ramp storage capacity

Control method	Ramp storage capacity (veh.)					
	20	40	60	80		
ALINEA\Q	23.19	24.26	26.82	27.45		
MS-FLC	25.74	26.96	27.98	28.77		

Table 10 summarizes the Δ_{MS} for the three simulation time scenarios. The figure indicates that in both control methods, the benefits in the mean speed are highest in the simulation 30-60-30 (31.93% and 37.47% for ALINEA\Q and FLC respectively), followed by the simulation 30-60-60. The evaluation times for the three scenarios are 75/90, 105/120, and 135/150 minutes respectively (excluding 15-minute warm-up period), and the ratios of the incident and non-incident period in the evaluation period are 30/45 (0.67), 60/45 (1.33), and 60/75 (0.80), respectively. This indicates that when the ratio of the incident and non-incident period is higher, the improvement in the mean speed of the control algorithms over No control increases. This coincides with the findings in cases a) and b) that the effectiveness of control is higher in the more critical mainline conditions.

615

Table 10. $\Delta_{\rm MS}$ versus the simulation period

Control method	S	imulation period (r	nin.)
Control method	30-30-30	30-60-30	30-60-60
ALINEA\Q	24.90	31.93	26.40
FLC	27.47	37.47	32.14

616 Discussions: feasibility and limitations

The study of results from the simulation scenarios shows that the benefits of control intervention (*ALINEA* and *FLC*) depend on the magnitude of traffic demand and incident situation. Broadly speaking, more significant benefits can be achieved under high traffic demands and critical incident conditions than under favourable conditions. The study of results from the sensitivity analysis provides further understanding on how the control performances change with changes in the input parameters, specifically:

- The benefits of control intervention are highly sensitive to the length of the network, in particular to the length of
 the upstream section. In general, the superiorities of the control methods over *No control* deteriorate as the network
 length increases.
- The superiorities of the control methods are less sensitive to the ramp storage capacity, in comparison to the network length. In general, the benefits of the control methods increase as the ramp storage capacity increases.
- The level of out-performance of the control algorithms is subject to the temporal structure of the simulation: when the ratio of the incident and non-incident period increases, the benefits over "No control" increase.
- It should be noted that the aforementioned findings are obtained from the model evaluation that was performed on a simplified network with an onramp, upstream and downstream incident segments, and a segment upstream of the ramp, under the local control as stated in the research scope. Although the model properties were further explored through sensitivity analysis with variations in the network length and simulation parameters, they are not verified for a more complicated network such as a corridor-wide control.

It should also be noted that there are no clear cuts between the terms *low*, *medium*, and *high* demands. They are loosely defined based on traffic demand in association with the reduced capacity. The question "to what range each of the demand categories covers" has not been verified numerically. An inspection of daily traffic volume profiles in the PIE's database revealed that *low-medium* demand level is usually associated with nighttime, while *medium-high* and *high* demands can mostly be observed in the daytime. Therefore, the MS-FLC has opportunities for practical applications in most of the time domain (daytime) when control intervention should be in operation.

640

Notwithstanding the important operational advantages, the MS-FLC has a number of limitations:

- The MS-FLC is complex and operationally expensive. It employs a considerable number of input parameters, thus
 extensive observations and measurements from the network are required.
- The essence of the fuzzy MS-FLC is the fuzzy rule base that formulates rules following fuzzy logic concept. In
 fuzzy logic, the input parameters are represented by fuzzy terms that are normally ill defined. In some cases, the
 partition of fuzzy sets must rely purely on personal judgements or common sense reasoning without having
 reference data to justify them based on solid technical grounds.
- The MS-FLC only enhances its performance if the rule base is well formulated with appropriate membership function design and input-output mapping. Otherwise, the system performance can deteriorate seriously.
- In calibrating parameters of membership functions of the fuzzy rule base, certain level of knowledge and expertise
 is required. The process of learning fuzzy rules requires a long time and the derivation of the membership functions
 can be tedious.
- In general, in the design of control system, stability analysis is one of the fundamental concerns. As an FLC, the
 MS-FLC is a highly non-linear system with complex stability behaviour. However, there exists no systematic
 methodology with respect to the stability analysis of the MS-FLC, to the best of the authors' knowledge.

655 Conclusion and Future Works

A multi-stage Fuzzy Logic Controller (MS-FLC) has been developed for traffic control under incidents on expressways. It 656 657 aims at assisting traffic operators in decision making on non-recurring congestion management in a systematic manner. The 658 decision-making process for traffic control during incidents on expressways include three tasks: (i) evaluation of incident traffic conditions, (ii) prediction of congestion tendency during the incident, and (iii) recommendation of local control 659 660 strategies and control actions to alleviate the congestion. Following this logic, a multi-stage composite structure is proposed. The MS-FLC is divided into three stages, each of which corresponds to one of the three tasks listed above, with rules being 661 executed sequentially from one stage to the next. The MS-FLC performance is evaluated by comparing with no control 662 663 scenario and ALINEA\Q, a popular local ramp control algorithm. Principal performance evaluation criteria include travel

time, waiting time on-ramp, total travel distance, mean speed, mean density, and queue length. The experiment evaluated the control algorithms under various traffic demand levels and incident scenarios. The experiment results show that in general MS-FLC outperforms ALINEA\Q with respect to global objectives. In particular, while the ALINEA\Q algorithm gives control preferences to the mainline, the MS-FLC algorithm gains a better balance between the mainline and the ramp.

In summary, the findings from this research allow the following conclusions to be drawn:

- The MS-FLC provides a systematic procedure in deriving control actions. Through the systematic assessment of
 prevailing traffic conditions in advance of control actions, the MS-FLC ensures that salient-influencing factors can
 be considered for proper control actions.
- For incident management, many types of data and information need to be gathered and analyzed, which may overload the traffic control operators. The MS-FLC resolves this challenging problem by its data-handling capability and knowledge representation to deliver simplified linguistic expression that is easy to understand by the operators.
- Flexibility of the performance: unlike ALINEA (ALINEA\Q) whose control algorithm does not consider incident situation, MS-FLC is specifically designed for incident management. Issues such as capacity reduction and queue management are addressed. However, MS-FLC can also be applied for recurring congestion management since the problem-solving strategy for both types of congestion aims at demand-capacity balance on the mainline and the ramp.
- The findings of this study have the extended potential for future research on application development of an adaptive MS-FLC. First, a MS-FLC with an adaptation component where parameters can be calibrated and rules can be modified on-line is worth exploring; second, effort should be extended to integrating the SVM short-term traffic prediction component for MS-FLC online operation; and third, future research should be devoted to development of the rule base and calibration of the MS-FLC model, as applicable for corridor-wide control.

685 Data Availability Statement

Some or all data, models, or code used during the study were provided by a third party. Direct requests for these materials
may be made to the provider as indicated in the Acknowledgements.

688 Acknowledgments

The authors would like to gratefully acknowledge the Land Transport Authority of Singapore for its provision of data usedin this study.

691 **References**

- Ali, F., D. Kwak, P. Khan, S. R. Islam, K. H. Kim, and K. S. Kwak. 2017. "Fuzzy ontology-based sentiment analysis of transportation
 and city feature reviews for safe traveling". *Transp. Res. Part C: Emerging Technol.* 77, 33-48.
- Arslan, T. C., and C. J. Khisty. 2005. "A rational reasoning method from fuzzy perceptions in route choice". *Fuz. Sets and Syst.* 150, 419-435.
- Bhandari, K. S., and G. H. Cho. 2019. "A resource oriented route selection framework using contextual information based on fuzzy
 logic". *Electr.* 2019 (8), 1023, doi:10.3390/electronics8091023.
- Cai, L., Q. Chen, W. Cai, X. Xu, T. Zhou, and J. Qin. 2018. "SVRGSA: A hybrid learning based model for short-term traffic flow
 forecasting". *IET Inte. Tran. Syst.* DOI: 10.1049/iet-its.2018.5315.
- Chen, Y. Y., Y. Cheng, and G. L. Chang. 2021. "Lane group-based traffic model for assessing on-ramp traffic impact". *J. Transp. Eng.*,
 Part A: Systems, 147(2), 04020152.
- Chowdhury, S., and M. O'Sullivan. 2018. "A fuzzy logic-genetic algorithm approach to modelling public transport users' risk-taking
 behavior". *Transp. Plan. and Tec.*, 41(2), 170-185.
- 705 Collotta, M., L. L. Bello, and G. Pau. 2015. "A novel approach for dynamic traffic lights management based on Wireless Sensor
- Networks and multiple fuzzy logic controllers". *Exp. Sys. with App.* 42(13), 5403-5415, doi:
- 707 <u>http://dx.doi.org/10.1016/j.eswa.2015.02.011</u>.
- Dhulipala, S., A. S. Kedia, P. S. Salini, and B. K. Katti. 2017. "Building a neuro-fuzzy based route choice model in metropolitan context:
 Surat city in India". *Transp. Res. Pro.* 25 (2017) 3203–3219.
- 710 Ge, Y. 2014. "A two-stage fuzzy logic control method of traffic signal based on traffic urgency degree". Mod. and Sim. in Eng., Article
- 711 ID 694185, 6pp, http://dx.doi.org/10.1155/2014/694185.
- Hashemi, H., and K. Abdelghany. 2018. "End-to-end deep learning methodology for real-time traffic network management". *Comp.-Aid Civ. and Inf. Eng.*, 33(10), 849-863.
- Hatri, E. C., and J. Boumhidi. 2018. "Fuzzy deep learning based urban traffic incident detection". Cog. Sys. Res. 50, 206-213.
- Hawas, Y. E., M. Sherif, and M. D. Alam. 2019. "Optimized multistage fuzzy-based model for incident detection and management on
 urban streets". *Fuz. Sets and Syst.*, 381, 78-104.
- 717 Kalinic, M., and J. M. Krisp. 2019. "Fuzzy inference approach in traffic congestion detection". Annals of GIS, 25(4), 329-336.
- 718 Imprialou, M. I. M., M. Quddus, and D. E. Pitfield. 2014. "High accuracy crash mapping using fuzzy logic". *Transp. Res. Part C:* 719 *Emerging Technol.* 42, 107-120.
- Lawrence W. L, and Y. C. Huang. 2006. "A rolling-trained fuzzy neural network approach for freeway incident detection".
- 721 *Transportmetric*, 2(1), 11-29, doi: 10.1080/18128600608685653.
- Lidbe, A. D., E. G. Tedla, A. M. Hainen, and S. L. Jones Jr. 2019. "Feasibility assessment for implementing adaptive traffic signal
- 723 control". J. Transp. Eng., Part A: Systems, 145(2), 05018002.

⁶⁹²

- Luan, X., Y. Wang, B. De Schutter, L. Meng, G. Lodewijks, and F. Corman. 2018. "Integration of real-time traffic management and train
- 725 control for rail networks Part 1: Optimization problems and solution approaches". *Transp. Res. Part B: Methodol.*, 115, 41-71.
- Luo, C., C. Huang, J. Cao, J. Lu, W. Huang, J. Guo, and Y. Wei. 2019. "Short-term traffic flow prediction based on least square support
 vector machine with hybrid optimization algorithm". *Neur. Proc. Lett.* 50: 2305-2322.
- Ma, J., B. L. Smith, and X. Zhou. 2016. "Personalized real-time traffic information provision: Agent-based optimization model and
 solution framework". *Transp. Res. Part C: Emerging Technol.* 64, 164-182.
- Memon, A., M. Meng, Y. D. Wong, S. H. Lam. 2015. "Rule-based mode choice model: INSIM expert system". J. of Transp. Eng.,
 141(4), 04014088.
- Memon, A., M. Meng, Y. D. Wong, and S. H. Lam. 2016. "Calibration of a rule-based intelligent network simulation model". *J. of Modern Transp.*, 24(1), 48-61.
- Motamed, M. 2016. "Developing a real-time freeway incident detection model using machine learning techniques". PhD Thesis. The
 University of Texas at Austin.
- 736 Motamed, M., and R. Machemehl. 2014. "*Real time freeway incident detection*", Technical Report No. SWUTC/14/600451-00083-1.
- 737 Center for Transportation Research, University of Texas at Austin.
- Nakatsuyama, M., N. Nagahashi, and N. Nishizuka. 1983. "Fuzzy logic phase controller for traffic functions in the one-way arterial
 road". In *IFAC 9th Triennial World Congress*. pp. 2865-2870.
- Pandey, S., P. Mathur, and T. Patil. 2017. "Real Time Traffic Signal Control using Fuzzy Logic Controller: REVIEW". *IEEE* 978-15090-4264-7/17.
- Papageorgiou, M., H. Hadj Salem, and J. M. Blosseville. 1991. "ALINEA: A Local Feedback Control Law for On-Ramp Metering".
 Transp. Res. Rec. 1320, pp. 58-64.
- Papageorgiou, M., C. Diakaki, V. Dinopoulou, A. Kotsialos, and Y. Wang. 2003. "Review of Road Traffic Control Strategies."
 Proceedings of the IEEE 96 (12): 2043–2067.
- Pappis C. D., and E. H. Mamdani. 1977. "A fuzzy logic controller for a traffic junction". *IEEE Trans. on Sys. Man and Cybe*. SMC-7,
 pp. 707-717.
- Peeta, S., and H. S. Mahmassani. 1995. "Multiple user classes real-time traffic assignment for online operations: A rolling horizon
 solution framework". *Transp. Res. Part C: Emerging Technol.* 3(2), pp. 83-98.
- Sasaki, T., and T. Akiyama. 1987. "Fuzzy on-ramp control model on urban expressway and its extension". *Transp. and Traf. Theory*. pp.
 377-395.
- 752 Sasaki, T., and T. Akiyama. 1988. "Traffic control process of expressway by fuzzy logic". Fuz. Sets and Syst. 26, pp. 165-178.
- 753 Simoni, M. D., and C. G. Claudel. 2017. "A fast simulation algorithm for multiple moving bottlenecks and applications in urban freight
- traffic management". Transp. Res. Part B: Methodol., 104, 238-255.
- Smaragdis, E., and M. Papageorgiou. 2003. A series of new local ramp metering strategies. In *Transportation Research Board* 82nd
 Annual Meeting. Washington, DC: Transportation Research Board.
- 757 Tariq, M. T., A. Massahi, R. Saha, and M. Hadi. 2020. "Combining machine learning and fuzzy rule-based system in automating signal
- timing experts' decisions during non-recurrent congestion". Transp. Res. Rec., 2674(6), 163-176.

- 759 Taylor, C., and D. Meldrum. 2000. Evaluation of a fuzzy logic ramp metering algorithm: a comparative study between three ramp
- 760 *metering algorithms used in the Greater Seattle area*. Technical report No. WA-RD 481.2. Washington State Department of
- 761 Transportation, USA.
- Toan, T. D., and S. H. Lam. 2005. "Development of a rule-based system for congestion management". In *Transportation Research Board 84th Annual Meeting*. Washington, DC: Transportation Research Board.
- Toan, T. D. 2008. "Development of a fuzzy knowledge-based system for local traffic control for incident management". PhD Thesis.
 School of Civil & Environmental Engineering, Nanyang Technological University.
- 766 Toan, T. D., and V. H. Truong. 2021. "Support vector machine for short-term traffic flow prediction and improvement of its model
- training using nearest neighbor approach". *Transp. Res. Rec.* 2675(4) 362–373.
- Toan, T. D., and Y. D. Wong. 2021. "Fuzzy logic-based methodology for quantification of traffic congestion". *Physica A: Statis. Mec. and its App.*, 570, 125784.
- Wang, X., Z. Ning, X. Hu, L. Wang, B. Hu, J. Cheng, and V. C. Leung. 2018. "Optimizing content dissemination for real-time traffic
 management in large-scale internet of vehicle systems". *IEEE Trans. on Veh. Tech.*, 68(2), 1093-1105.
- Xiao, J., X. Gao, Q. J. Kong, and Y. Liu. 2013. "More robust and better: A multiple kernel support vector machine ensemble approach
 for traffic incident detection." *Jour. of Adva. Tran.*, 858-875.
- Xu, J., X. Zhao, and D. Srinivasan. 2013. "On optimal freeway local ramp metering using fuzzy logic control with particle swarm
 optimization". *IET Intell. Transp. Syst.*, 2013, Vol. 7, Iss. 1, pp. 95–104, doi: 10.1049/iet-its.2012.0087.
- Li, Y., and W. Xu. 2018. "Short-term traffic flow forecasting based on SVR". Adva. in Engi. Rese., 166: 57-61.
- 777 Yusupbekov, N. R., A. R. Marakhimov, H. Z. Igamberdiev, and Sh. X. Umarov. 2015. "An adaptive fuzzy-logic traffic control system in
- conditions of saturated transport stream". *The Scien. Wor. J.* Article ID 6719459, 9pp, <u>http://dx.doi.org/10.1155/2016/6719459</u>.
- Zaied, A. N. H., and W. Al Othman. 2011. "Development of a fuzzy logic traffic system for isolated signalized intersections in the State
 of Kuwait". *Exp. Sys. with App.* 38(8), 9434-9441.
- Zhan, L., and P. D. Prevedouro. 2011. "User perceptions of signalized intersection level of service using fuzzy logic". *Transportmetrica*,
 7(4), 279–296.
- Zhang, M., T. Kim, X. Nie, and W. Jin. 2001. *Evaluation of On-ramp Control Algorithms, California PATH*. Research Report, UCB ITS-PRR-2001-36.
- Zhao, J., W. Ma, Y. Liu, and K. Han. 2016. "Optimal operation of freeway weaving segment with combination of lane assignment and
 onramp signal control". *Transportmetrica A: Trans. Sci.* DOI: 0.1080/23249935.2016.1146927
- 787 Zhong, R.X., A. Sumalee, T. L. Pan, and W. H. K. Lam. 2014. "Optimal and robust strategies for freeway traffic management under
- demand and supply uncertainties: an overview and general theory". *Transp. A: Transp. Sci.*, 10:10, 849-877, DOI:
- 789 10.1080/23249935.2013.871094.